Author's response to reviews

Title: Evaluation of a self-management programme for congestive heart failure patients: design of a randomised controlled trial [ISRCTN88363287]

Authors:

Esther STF Smeulders (e.smeulders@zw.unimaas.nl)
Jolanda CM van Haastregt (j.vanhaastregt@zw.unimaas.nl)
Elisabeth FM van Hoef (l.vanhoef@zw.unimaas.nl)
Jacques THM van Eijk (j.vaneijk@zw.unimaas.nl)
Gertrudis IJM Kempen (g.kempen@zw.unimaas.nl)

Version: 2 Date: 4 July 2006

Author's response to reviews: see over
Dear editors,

Enclosed you will find the paper “Evaluation of a self-management programme for congestive heart failure patients: design of a randomised controlled trial” [ISRCTN88363287] which we would like to submit for publication in BMC Health Services Research. We state that the content of this manuscript has not been submitted for publication to any other journal.

The manuscript presents the study protocol of a multi-centre randomised controlled trial, evaluating the long-term effectiveness of a structured self-management programme (Chronic Disease Self-Management Programme, Lorig et al.) among congestive heart failure outpatients in the Netherlands. The manuscript is innovative as it concerns the first evaluation study of the programme specific among congestive heart failure patients. In addition, this cognitive-behavioural intervention emphasizes the patients’ central role and responsibility in managing their illness, which seems crucial for dealing with congestive heart failure in an acceptable way.

This trial has received a grant from a major funding body; the Netherlands Heart Foundation (project number: 2002B005). Approval to conduct this trial was granted by the Medical Ethics Committee of Maastricht University/ University Hospital Maastricht responsible for supervision of the study (MEC number: 03-231-5). Local feasibility of the study design was additionally assessed in each other participating hospital as well. All significant documents concerning the ethics approval and funding procedures are available.

We hope that you will consider our paper for publication in BMC Health Services Research.
On behalf of the co-authors,

sincerely,

Esther S.T.F. Smeulders, MSc
Maastricht University, Faculty of Health Sciences
Dep. of Health Care Studies
P.O. Box 616
6200 MD Maastricht
The Netherlands

+31 43 388 41 83
E.Smeulders@ZW.unimaas.nl